Author's response to reviews

Title: Treatment of refractory epilepsy with natalizumab in a patient with multiple sclerosis. Case report

Authors:

Stefano Sotgiu (stesot@hotmail.com)
Maria R Murrighile (cineuro@uniss.it)
Gabriela Constantin (gabriela.constantin@univr.it)

Version: 4 Date: 26 August 2010

Author's response to reviews: see over
Dear Editor,

Please find attached our revised case report entitled: "Treatment of refractory epilepsy with natalizumab in a patient with multiple sclerosis. Case report".

All revisions requested were carefully addressed and the modifications made to the text are, in detail:

1. Major revisions:
   a. A conclusion section has been inserted after the Discussion
   b. The volume of text in the legend to figure 2 has been reduced from 397 to 257 words. The word count of figure legend 1 is 186, already below the suggested limit of 250.
   c. The Table title and Table legend has been moved to the main text. In particular, they have been adapted and pasted at the beginning of the Case Presentation section.

2. Minor revision:
   a. All text is in black
   b. Additional files have been removed
   c. Titles of the figures have been deleted
   d. White space around the figures has been minimized
   e. Typographic errors have been corrected

We hope the manuscript is now suitable for publication in BMC Neurology.

We thank you for your consideration.

For the authors

Stefano Sotgiu and Gabriela Constantin